Author’s response to reviews

Title: Strategies for Evaluating Self-Efficacy and Observed Success in the Practice of Yoga Postures for Therapeutic Indications: Methods from a Yoga Intervention for Urinary Incontinence among Middle-aged and Older Women

Authors:

Francesca Nicosia (Francesca.Nicosia@ucsf.edu)
Lisha Nadra (Nadra.Lisha@ucsf.edu)
Margaret Chesney (Margaret.Chesney@ucsf.edu)
Leslee Subak (lsubak@stanford.edu)
Traci Plaut (Traci.Plaut@ucsf.edu)
Alison Huang (Alison.Huang@ucsf.edu)

Version: 1 Date: 14 May 2019

Author’s response to reviews:

As requested, we have revised the title of the manuscript and text in the introduction to clarify that the study results are not clinical trial outcomes but previously unpublished methods for evaluating self-efficacy and observed competency in performing yoga postures.